Author's response to reviews

Title: The optimization of the diagnostic work-up in patients with suspected obstructive lung disease

Authors:

Frank J Visser (f.visser@cwz.nl)
Milena JMM van der Vegt (milena.roorda@hetnet.nl)
Gert Jan van der Wilt (G.vanderWilt@mta.umcn.nl)
Julius P Janssen (j.janssen@cwz.nl)

Version: 7 Date: 20 November 2010

Dear editor,

Final revision of the accepted paper: Optimization of the diagnostic work-up in patients with suspected obstructive lung disease.

Point by point reply:
Formatting changes to be made:

--------------------

Major revisions (we require the author to make these changes)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.
OK DONE

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Methods, Results, Conclusions
OK DONE

Tables- please remove the visible vertical lines from your tables
OK DONE

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Introduction: Please rename this section 'Methods'.
OK DONE

Table: Please include the title above the table.
OK DONE

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our
online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

Figure 1 is cropped OK DONE

Additional file: Please remove this.
Ok done

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Ok.


To submit your revised manuscript

Sincerely,

Frank Visser MD